Author’s response to reviews

Title: Agreement Study between Color and IR Retinal Images Based on Retinal Vasculature Morphological Parameters

Authors:
Aqsa Ajaz (aqsa.ajaz@gmail.com)
Behzad Aliahmad (behzad.aliahmad@rmit.edu.au)
Himeesh Kumar (himeesh@gmail.com)
Marc Sarossy (marc@sarossy.com)
Dinesh Kumar (dinesh@rmit.edu.au)

Version: 2 Date: 29 Nov 2018

Author’s response to reviews:

we have made the required changes and listed below:

1. Emails of all authors

2. Methodology to Methods

3. 'Ethics Approval' heading to 'Ethics approval and Consent to Participate'.

4. Details of the contribution of co-author, HK.

Thank you, Dinesh